Author’s response to reviews

Title: The role of MRI and clinicopathologic features in predicting the invasive component of biopsy-confirmed ductal carcinoma in situ

Authors:

Ga Young Yoon (3770ghwo@hanmail.net)
Woo Jung Choi (wjc@amc.seoul.kr)
Joo Hee Cha (jhcha@amc.seoul.kr)
Hee Jung Shin (docshin@amc.seoul.kr)
Eun Young Chae (chaeey@hanmail.net)
Hak Hee Kim (hhkim@amc.seoul.kr)

Version: 1 Date: 26 Apr 2020

Author’s response to reviews:

We additionally changed mauscript formatting as your recommendation.

• Please add a separate “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.
  -> Thank you for your recommendation. We separated "Conclusions" section.

• Abbreviations should be after Conclusions and before Declarations section.
  -> Thank you for your recommendation. We placed "Abbreviations after conclusions.

• Please note that all manuscripts must contain all the following sections under the heading 'Declarations' which should be before the References. Provide all sections, rename headings and arrange the order as indicated below:
  -> Thank you for your recommendation. We renamed all sections of "Declarations"

• When full names are used in AC section
  Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. For example, the initials of John Smith would be ‘JS’. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors, for example, by adding their middle name initial.
  -> Thank you for your recommendation. We changed author's names to full initials.
1. As indicated by our submission guidelines, please format your manuscript to include the following subheadings: Background, Methods, Results, Discussion, and Conclusions.

-> Thank you for your comment. We revised the manuscript's format as following subheadings: Background, Methods, Results, Discussion, and Conclusions.

2. Please provide a 'Declarations' section for your manuscript before the references section. Please ensure that this section contains all of the sub-sections as described by our submission guidelines:

-> Thank you for your comment. We made 'Declarations' section.

3. We note you have included the Ethics approval and consent to participate statement in your title page. Please move this information under Declarations under the Ethics approval and consent to participate heading.

-> Thank you for your comment. We made 'Declarations' section.

4. Please provide a List of abbreviations after the Conclusions section. When abbreviations are used in the text, they should be defined in the text at first use and included in this list.

-> Thank you for your comment. We made the list of abbreviations.